Author's response to reviews

Title: Vav3 oncogene activates estrogen receptor and its overexpression may be involved in human breast cancer.

Authors:

Kiwon Lee (leeko@UCMAIL.UC.EDU)
Yin Liu (yin.liu@uc.edu)
Jun Qin Mo (Jun.Mo@cchmc.org)
Jinsong Zhang (zhangj5@ucmail.uc.edu)
Zhongyun Dong (dongzu@ucmail.uc.edu)
Shan Lu (shan.lu@uc.edu)

Version: 4 Date: 22 May 2008

Author's response to reviews: see over
Dr. Melissa Norton  
Editor-in-Chief  
BMC Cancer

Dear Dr. Norton,

We are submitting the revised version of our manuscript, entitled “Vav3 oncogene activates estrogen receptor and its overexpression may be involved in human breast cancer” (MS: 1236477421179147), for publication in *BMC Cancer*. Formatting of the manuscript has been changed according for online production process. These changes include affiliations, email address, references, figure number in the image, and figure cropping issue.

I appreciate very much that BMC cancer is interested in my research.
Best regard.

Shan Lu